CrossMark

Assessing the Relationship between Vector Indices and Dengue Transmission: A Systematic Review of the Evidence

Leigh R. Bowman¹, Silvia Runge-Ranzinger², P. J. McCall¹*

1 Liverpool School of Tropical Medicine, Liverpool, United Kingdom, 2 The Special Programme for Research and Training in Tropical Diseases of the World Health Organization (WHO/TDR), Geneva, Switzerland

Abstract

Background: Despite doubts about methods used and the association between vector density and dengue transmission, routine sampling of mosquito vector populations is common in dengue-endemic countries worldwide. This study examined the evidence from published studies for the existence of any quantitative relationship between vector indices and dengue cases.

Methodology/Principal Findings: From a total of 1205 papers identified in database searches following Cochrane and PRISMA Group guidelines, 18 were included for review. Eligibility criteria included 3-month study duration and dengue case confirmation by WHO case definition and/or serology. A range of designs were seen, particularly in spatial sampling and analyses, and all but 3 were classed as weak study designs. Eleven of eighteen studies generated *Stegomyia* indices from combined larval and pupal data. Adult vector data were reported in only three studies. Of thirteen studies that investigated associations between vector indices and dengue cases, 4 reported positive correlations, 4 found no correlation and 5 reported ambiguous or inconclusive associations. Six out of 7 studies that measured Breteau Indices reported dengue transmission at levels below the currently accepted threshold of 5.

Conclusions/Significance: There was little evidence of quantifiable associations between vector indices and dengue transmission that could reliably be used for outbreak prediction. This review highlighted the need for standardized sampling protocols that adequately consider dengue spatial heterogeneity. Recommendations for more appropriately designed studies include: standardized study design to elucidate the relationship between vector abundance and dengue transmission; adult mosquito sampling should be routine; single values of Breteau or other indices are not reliable universal dengue transmission thresholds; better knowledge of vector ecology is required.

Citation: Bowman LR, Runge-Ranzinger S, McCall PJ (2014) Assessing the Relationship between Vector Indices and Dengue Transmission: A Systematic Review of the Evidence. PLoS Negl Trop Dis 8(5): e2848. doi:10.1371/journal.pntd.0002848

Editor: Amy C. Morrison, University of California, Davis, United States of America

Received December 9, 2013; Accepted March 27, 2014; Published May 8, 2014

Copyright: © 2014 Bowman et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: The study was funded by EU grant FP7-281803 IDAMS (www.idams.eu). The contents of this publication are the sole responsibility of the authors and do not necessarily reflect the views of the European Commission. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: mccall@liv.ac.uk

Introduction

Global dengue incidence has increased markedly over the past 50 years to the point where it is now the most widespread mosquito-borne arboviral disease. The World Health Organisation (WHO) has estimated that 50-100 million dengue infections occur annually, while a recent study calculated that the true figure may be closer to 400 million [1–3]. Dengue is endemic throughout the tropics, and almost half of the world's population are at risk of infection, 75% of whom live in the Asia-Pacific region [4]. Dengue has been confirmed in 128 countries worldwide [4,5] with major social and economic consequences [6–10].

Dengue is transmitted by *Aedes* mosquitoes, primarily by the highly urban-adapted vector *Aedes aegypti*, and a secondary vector *Aedes albopictus* [11]. *Ae. aegypti* thrives in the man-made urban environment, particularly in deprived communities where water storage is routine, sanitation is poor and non-biodegradable containers accumulate.

The abundance of dengue vectors species as well as dengue transmission generally show seasonal variation. Depending on the local ecology, these patterns can be in part driven by meteorological parameters such as rainfall and temperature [12,13]. Vector surveillance is recommended by WHO and is a routine practice in many dengue-endemic countries to provide a quantifiable measure of fluctuations in magnitude and geographical distribution of dengue vector populations, ultimately with the purpose of predicting outbreaks and evaluating control [14]. The standard protocol relies on the Stegomyia indices, which sample the immature mosquito stages (larvae and pupae) alone [15]. This approach was developed over 90 years ago [16] for yellow fever, a markedly different infection (zoonotic in origin though ultimately transmitted between humans by Ae. aegypti) during a very different era (i.e. in terms of urbanization levels and human population densities). Focks (2004) questioned the reliability and sensitivity of the Stegomyia indices because they correlate poorly with abundance of adult

Author Summary

Routine sampling of mosquito vector populations is common in dengue-endemic countries worldwide despite doubts about methods used or the correlation between vector density and dengue transmission. This systematic review examined the published evidence investigating associations between vector indices and dengue cases. From a total of 1205 papers identified in database searches, 18 were included for review. A range of designs were seen, particularly in spatial sampling and analyses, and all but 3 were classed as weak study designs. Thirteen studies investigated associations between vector indices and dengue cases: 4 reported positive correlations, 4 found no correlation and 5 reported ambiguous/unreliable associations. Of 7 studies that measured the Breteau Index, 6 reported dengue transmission at levels below the currently accepted threshold of 5. There was little evidence of guantifiable associations between vector indices and dengue transmission that could reliably be used to predict outbreaks. Furthermore, appropriately designed studies are required to elucidate the relationship between vector abundance and dengue transmission. Recommendations include: standardizing study designs, particularly with respect to spatial heterogeneity; vector surveillance programs should sample adult mosquitoes; global values of the Breteau Index are not reliable universal dengue transmission thresholds; and better knowledge of vector ecology is required.

mosquitoes, (*i.e.* the actual vector stage) which should be sampled directly [15]. Focks and others recommended sampling adult mosquitoes directly or indirectly via pupal/demographic surveys (calculating a pupae per person/area index, defined as the number of pupae divided by the number of residents/area surveyed) [15,17]. Indices based on actual counts of adult female *Ae. aegypti* infesting houses are likely to be the most accurate, but this is rarely done [15].

The Stegomvia indices remain central to the monitoring of dengue vector populations. The most commonly used indices are the House (or 'premise') index (HI - percentage of houses infested with larvae and/or pupae;) the Container index (CI - percentage of waterholding containers infested with larvae and/or pupae) and the Breteau index (BI - number of positive containers per 100 houses inspected) [14]. Variations in sampling protocols are common and can lead to significant variations in indices: e.g. sampling may be carried out indoors or outdoors only, or at both locations; the presence of cryptic breeding sites may lead to under-sampling or complete omission of certain sites; failure to distinguish Aedes aegypti/ albopictus from other common mosquito species, or from each other, may lead to overestimates. Little is known about the relationship between differing proportions of the various sampled larval instars and the accuracy of these data as proxy measures of adult mosquito abundance [17]. Finally, although ovitraps (water-filled pots in which Aedes aegypti lay their eggs) are widely used as a simple sampling tool, Focks [15] showed very convincingly that their reliability is limited to indicating vector presence or absence.

Despite these doubts, many dengue control authorities worldwide routinely collect vector population data based on these indices, although the mathematical relationship between any of the indices and dengue transmission is far from clear. Thresholds indicating dengue outbreak risk for House and the Breteau indices (HI = 1%, BI = 5) have been used for many years [18,19], even though these values were developed for yellow fever many decades earlier. Simple thresholds may be valid in some situations [20], but a universal critical threshold applicable across many contexts, has never been determined for dengue. In pursuing the goal of identifying dengue thresholds, Scott & Morrison [21] defined the fundamental knowledge gaps as: 1) what is an acceptable level of dengue risk?; 2) what are the mosquito densities necessary to achieve that goal?; 3) what is the best way to measure entomological risk?; 4) at what geographic scale are the components of dengue transmission important? While a number of mathematical models have explored the value of thresholds or rates of change in the vector population for the prediction of dengue outbreaks [22,23], these knowledge gaps remain and continue to hinder progress [24]. For convenience, dengue outbreaks are often defined as periods when dengue incidence is equivalent to the mean plus 2 standard deviations during the same month of the previous year [25].

Effective dengue surveillance and early warning systems, using information from multiple epidemiological sources, are an important goal for numerous countries worldwide. To determine the value of vector surveillance for such systems, the findings of a systematic review examining the evidence for a relationship between mosquito indices and dengue cases are reported here.

Methods

Objectives

The aim of the study was to evaluate the potential value of vector or entomological survey data for dengue surveillance by examining the evidence from studies that investigated quantitatively the relationship between vector indices and dengue cases. The specific objectives were:

- 1. To identify vector surveillance methods and indices used for the routine monitoring of *Aedes aegypti* or *Aedes albopictus* populations in any geographic location.
- 2. To examine how entomological indices correlated with dengue incidence.
- To examine the effectiveness or accuracy of vector surveillance in predicting dengue outbreaks and consider how this might be improved.

Search Strategy

A review protocol was established and agreed upon by all authors. Guidelines from the Cochrane Handbook for Systematic Reviews and the PRISMA Group were followed as standard methodologies [26,27]. The databases WHOLIS, PubMed, EMBASE, LILACS and Web of Science were searched using the Medical Subject Heading (MeSH) "dengue" followed by the Boolean operator "and" combined with one of each of the following 'free text' terms in succession: 'entomological surveillance', 'oviposition trap', 'house index', 'container index', 'Breteau index', 'pupal index', 'pupal survey', 'adult collection', 'sticky trap', 'aspirator collection', 'resting collection', 'landing collection', 'vector density'. The reference list of each of the included studies was also searched, and "grey literature" was sought by communication with authors for cited unpublished documents.

Results were collated in EndNote (EndNote X5, Build 7473) where abstracts were reviewed in accordance with agreed inclusion and exclusion criteria. Full text review was completed using 'Papers' (Papers 2, version 2.2.10). No limits were placed on year of publication, language or location.

Inclusion and Exclusion Criteria

The criteria for inclusion or exclusion of individual studies were set in advance (Table 1) and were used to assess each abstract and/ or the full text.

Table 1. Criteria for inclusion or exclusion of studies.

Inclusion Criteria	Exclusion Criteria
Any study where entomological surveillance of <i>Aedes</i> spp. was undertaken for >3 months (or for the duration of a dengue outbreak) in conjunction with number of reported dengue cases	Studies with only one outcome of interest (entomological surveillance OR dengue cases);
Any study type with all empirical data gathered within the same time period	Opinion papers; review articles; retrospective analyses comparing data generated at different time points
Confirmed and/or probable dengue cases identified using WHO standard case definition and/or serology	Qualitative dengue reports

doi:10.1371/journal.pntd.0002848.t001

	Potential relevant citations identified by the search of databases (n=1205; including duplications) (I) WHOLIS = 1 (II) LILACS = 57 (III) Pubmed = 457 (IV) Web of Science = 467 (V) EMBASE = 201 (VI) Unpublished data = 1 (VII) Manual Search = 21
1103 citations excluded after abstract review (including 684 duplications)	Studies retrieved for full text evaluation n=102 (I) WHOLIS = 0 (II) LILACS = 4 (III) Pubmed = 56 (IV) Web of Science = 19 (V) EMBASE = 15 (VI) Unpublished data = 1 (VII) Manual Search = 7
Studies excluded after initial evaluation of full text and application of inclusion and exclusion criteria from the systematic review (n = 84)	
	Studies included in systematic review n = 18

Figure 1. Search Tree. Diagram of searches performed and the number of articles returned and examined at each stage. doi:10.1371/journal.pntd.0002848.g001

Ref. Number	Study	<u> </u>	ature Vec	tor Indice	a.		Adult mosquitoes sampled	Egg (ovitrap) sampled	Location		Sample spatial unit	Significant (<i>p</i> ≤0.05) increase in vector indices recorded during dengue transmission
		Ū	Ŧ	В	BI _{max}	Pupal Index			Indoor	Indoor + Outdoor		
20	Sanchez <i>et al</i> , 2010.			•	-		1		>		Block; N'hood	1
29	Sanchez <i>et al.</i> , 2006.	-	•				+		>		Block; N'hood	+
30	Chadee, 2009		•			pupae/person	+			>	Premise	+
31	Pham <i>et al.</i> , 2011		•	-			+		>		Premise	+
32	Gurtler <i>et al.</i> , 2009		•				٤		>		N'hood; City	ł
33	Katyal <i>et al.</i> , 2003	٠	٠	٠			٤					ł
34	Chadee <i>et al.</i> , 2005						-/+			>	Premise*	-/+
35	Romero-Vivas & Falconar, 2005		•	-		pupae/premise	I			>	Premise	I
36	Foo <i>et al.</i> , 1985		٠	•			I		>		Premise	I
37	Sulaiman <i>et al.</i> , 1996		⊙	⊡			-/+		>		City zone	-/+
38	Honorio <i>et al.</i> , 2009						1	>			Premise	I
39	Rubio-Palis et al., 2011						+		>		Premise	+
40	Lin & Wen, 2011			•	•		-/+			`	District; Min admin unit	-/+
41	Chaikoolvatana <i>et al.</i> , 2007		•	-			ł			`	Village	ł
42	Chadee <i>et al.</i> , 2007		•	•			٤			`	County	ł
43	Correa <i>et al.</i> , 2005						-/+				District; Trial area	-/+
44	Fernandez <i>et al.</i> , 2005	٠	٠	٠			-/+				Premise	-/+
45	Arboleda <i>et al.</i> , 2012			٠			I				0.25 km²	I

Immature vector samples are denoted as: ◆ larvae only; ■ larvae and pupae; ⊙ *Aedes aegypti & Aedes albopictus* combined. Cells marked ✓ indicate the sampling activity was done. The Sample spatial unit referred to as 'P*' is the 'premise with cardinal points index' [34]; 'N'hood' = neighborhood. In the right-hand column, the reported association between vector indices and dengue cases is classed as: +' positive association; '+ -' ambiguous association; '~ inconclusive or weak association. Absence of any entries in cells indicates no data or information was reported. doi:10.1371/journal.pntd.0002848.t002

Definitions

The following definition was used for the term 'vector surveillance': "Any ongoing surveillance of entomological indices, including larval indices (House Index (HI), Container Index (CI), Breteau Index (BI)), pupal indices (Pupal Productivity Index (PPI) and other variations), oviposition trap data and data from adult mosquito collections (methods include sticky, traps, CO₂, odorbaited, visual or other traps, resting catches, human landing catches), used in relation to dengue outbreak/control."

Quality Assessment

Given the strict nature of the inclusion criteria, study design was assessed at the data extraction stage using the Quality Assessment Tool for Quantitative Studies (QATQS) [28]. QATQS provides a recognized standardized method to assess study quality by assigning scores based on possible selection bias, study design, confounders, data collection methods, intervention integrity and statistical analyses. This ensured each study could be ranked qualitatively. The study design classes were intervention, casecontrol and longitudinal. If clarification was required, authors were contacted for any missing data or information.

Data Extraction and Assessment

The information extracted included first author, year of publication, year of study, population size, study design, indices and case definitions, study objectives, duration of study, frequency of data collection, results and conclusions (as viewed by all reviewers; Table S1). A table of bias was created to help identify the strengths and weaknesses of each study (Table S2).

Ethics Statement

No ethical review was required for this systematic literature review.

Results

A total of 1205 potentially relevant studies were identified in the database search. After reviewing abstracts, 102 were selected and retrieved for full text evaluation, of which 18 were considered to have satisfied all inclusion and exclusion criteria and explored in detail (Figure 1) [20,29–45].

Regarding the 84 studies excluded, the most common reasons for exclusion were: study duration less than 3 months (22 studies); absence of a reliable dengue case definition (21 studies); use of datasets that did not correspond temporally or spatially (19 studies). Note that although such dislocated spatial comparisons were not captured by the exclusion criteria originally defined (simply because it had not been expected), exclusion at this point was considered to be valid. Other reasons for exclusion were: measurement of only one outcome (i.e. vector or dengue cases only: 9 studies); opinion or review articles (8 studies); use of incomplete datasets - where only 'selected' portions of all of the data available during the study period were used (5 studies). Again, although the latter reason was not captured by the original criteria, exclusion of studies where this occurred was considered to be valid. Full details of the 18 studies reviewed are summarised in the supporting data files (Checklist S1, Table S1, Table S2).

The origin of the data used in analyses differed between studies. Some generated novel data as an integral part of the study, thus ensuring complete or independent control over the quality of the data obtained, while others obtained existing or retrospective data from external sources, including local surveillance data (*e.g.* local government records, private companies, hospitals or health centers, independent physicians and self-reported data). Twelve studies generated vector data [30–32,34–40,42–44], five generated dengue case data [29,30,35,36,38], four of which generated both vector and dengue case data [30,35,36,38].

Study Design

Fourteen studies were longitudinal, two were case-control, one was an ecological study (as defined by the unit of analysis) and one was a vector control intervention. Applying QATQS [28], fifteen studies [20,30–33,35–41,43–45] scored 3 (defined as a weak study), two studies [29,42] scored 2 (a moderate study design) and one study [34] scored 1 (a strong study design)(Annex 2). In the latter study, Chadee and colleagues [34] used controls matched on age and sex from a neighboring community, although the report did not state whether or not this process was randomized.

Vector Sampling

Details of the sampling protocols used in each study are shown in Table 2. Eleven of eighteen studies generated indices for immature stages of the vector and collected combined larval and pupal numbers to calculate either the CI, HI or BI [20,29– 32,34,35,37,40,42,43]. One of these [37] combined *Ae. aegypti* and *Ae. albopictus* data. Four studies sampled only larvae [33,36,44,45].

Thirteen studies reported the location of the immature stage mosquito samples: six studies sampled both indoor and outdoor containers [30,34,35,40–42], while seven searched indoor containers only [20,29,31,32,36,37,39]. Thus, where reported, all studies included indoor sampling.

Pupal indices were reported in two studies [20,35]. Adult mosquitoes were sampled in three studies [38,39,43].

Relationship between Entomological Indices and Dengue Cases

Thirteen studies examined the association between entomological indices and dengue, using a range of different statistical approaches. Seven studies calculated regression coefficients [36,37,39,40,43–45], two calculated rate ratios [31,38], one calculated odds ratios [29] and two calculated the G-test for significance [32,36]. One study used only specificity, sensitivity and positive and negative predictive values [20].

The spatial unit of analysis, an important consideration in dengue epidemiology (see Discussion) varied considerably across studies, with units ranging from individual houses, housing blocks and clusters to neighborhoods and even large municipalities (Table 2).

Four studies reported statistically significant positive relationships between entomological indices and dengue incidence [29– 31,39]. Of these, only one sampled adult mosquitoes (33% of those studies that sampled adults) [39] while the remainder sampled immature stage mosquitoes (20% of all those that sampled immatures) [29,30,31](Table 2). These are discussed in detail here.

Evidence for Positive Correlation between Vector Indices and Dengue Cases

Sanchez (2006) [29] conducted a case control study using two geographical units for analysis, blocks (units of approximately 50 houses) and neighborhoods (each containing approximately 9 blocks). Any block or neighborhood with at least 1 confirmed case was considered positive, while a control was defined as a block or neighborhood without confirmed cases. HI and BI mean values were "consistently, substantially and significantly higher" in blocks with dengue cases compared with control units. An odds ratio (OR) of 3.49 (p<0.05) for dengue transmission was associated with the presence of a single positive container in a block; fifteen of the seventeen dengue cases recorded lived in a neighborhood where at least 1 block had a BI>4.

In Trinidad, Chadee (2009) [30] compared retrospective routine entomological household data with concurrent entomological data taken from confirmed dengue households, using a cardinal points approach (*i.e.* the 'index' house plus the four adjacent houses at its cardinal points). Chadee found that significantly more (P<0.001) immatures were collected during dengue case investigations than during the routine inspection and treatment cycles. The report also stated that pupae per person indices were higher and significantly more adults emerged (as a function of total pupae count collected from household containers) at locations where dengue was confirmed at the index house, compared with routine investigations.

Pham *et al.* [31], examined monthly dengue case data, vector larval indices and meteorological data from central Vietnam, between 2004 and 2008. They found significant associations between all entomological indices and dengue cases by univariate analysis but only the HI and "household mosquito index" (not defined in the paper), temperature and rainfall were significant after multivariate analysis.

In Venezuela, Rubio-Palis *et al.* [39] used a simple regression analysis to investigate correlations between vector indices, climatic variables and dengue incidence for the period 1997–2005. Analyses indicated a significant relationship ($\mathbb{R}^2 = 0.9369$) between the numbers of dengue cases, *Ae. aegypti* abundance (both immatures and adults) and rainfall. Acknowledging the retrospective nature of the study, the authors expressed caution in the predictive value of the findings. Moreover, another limitation was that entomological data were derived only from actual homes and neighbouring houses of confirmed dengue cases but no data were collected from 'control' houses.

Value of Vector Indices for Advance Warning of Dengue Outbreaks

Within these four studies was some additional evidence that observed changes in vector indices might be useful for the prediction of impending dengue transmission or outbreaks. In Cuba, Sanchez (2006) [29] reported that blocks with BImax (defined as the highest or 'maximum' block level BI in a neighborhood) values greater than 4 were significantly more likely to record positive cases in the following month, and had a 3-5 times greater dengue risk in comparison with control blocks. The report concluded that BImax>4 and neighborhood BI>1 during the preceding 2 months provided "good predictive discrimination". In northern Venezuela Rubio-Palis et al. [39] found the most significant correlation between rainfall levels and the appearance of dengue cases two months later, indicating that the magnitude of outbreaks might be predictable to some extent following periods of rainfall. Pham et al. [31] confirmed an association between dengue transmission and periods of higher rainfall and mosquito abundance in the central highlands of Vietnam, but did not indicate whether this could be used in advance of transmission as a predictive tool.

Unreliable or Absence of Correlation between Vector Indices and Dengue Cases

A further five studies [34,37,40,43,44] reported ambiguous evidence of associations, both positive and negative, between entomological data and dengue cases. In Belo Horizonte, Correa *et al.* [43] found a 5–7 fold increase in mean monthly dengue incidence where the 'infestation rate' (defined as house index) was "between 1.33% and 2.76% and equal to or higher than 2.77% when compared to areas showing 0.45% or less", although it was unclear whether or not this was statistically significant. They reported a moderate but significant correlation between adult *Aedes spp.* infestation rates and numbers of dengue cases ($\mathbf{R} = 0.67$) even



Figure 2. Range of Breteau indexes reported during dengue transmission. Dotted line indicates a BI value of 5, which has been considered a transmission threshold for dengue [21,45,52]. Note: Includes all data where available, whether statistically significant or insignificant. doi:10.1371/journal.pntd.0002848.g002

though HI and dengue cases were only weakly correlated (R = 0.25 at the municipal level; R = 0.21 and R = 0.14 at the district and village level). Sulaiman *et al.* [37] reported a significant correlation between BI and HI and dengue cases in certain areas of Kuala Lumpur, but not in others. In Trinidad, Chadee *et al.* [34] found that 75% of DHF cases were located in areas where BI was greater than 10, although BI and dengue infections were rarely correlated. An additional two studies reported either very low correlations between vector indices and dengue [44], or utilized highly variable inter-annual data precluding such analyses [40].

Four studies, from Malavsia [36], Brazil [35] and Colombia [35,45] found no statistically significant relationships between entomological indices and dengue cases. Foo et al. [36] observed a positive but non-significant association between dengue cases and HI and BI, which they suggested may have been influenced by the small sample size, the presence of Ae. albopictus and sociodemographic factors. Honorio et al. [38] found no significant associations between recent dengue cases and Ae. aegypti densities and proposed that infections received outside the home were responsible. In Colombia, Romero-Vivas and Falconar [35] reported distinct positive temporal correlations between the larval density index and pupal density index (p < 0.005) and a negative association between the larval density index and egg density index (p < 0.01); however, they found no correlation between any of the larval, pupal or adult indices with either rainfall or dengue-like cases. The spatial model of Arboleda et al. found no indication that the BI was in any way correlated with the dengue cases or those areas predicted as 'suitable' [45].

In the remaining studies [20,32,33,41,42] a variety of mixed, inconclusive or weak associations were reported. Gurtler et al. conducted analyses on the effect of a given intervention on mosquito indices but not on dengue cases [32]. Although Katyal et al. [33] did not present any statistical analysis, they reported the observation that over a five year period, a fall in cases was visually correlated with a fall in indices. However, they conceded that "an increasing trend of cases was observed [in 2001] in spite of a further declining HI trend", and concluded that HI had no predictive value at the 'macro' level. Despite the absence of statistical analysis, Chaikoolvatana et al. [41] reported a suggestive link between dengue haemorrhagic fever (DHF) during peak annual rainfall months and high abundance of mosquitoes. Chadee et al. observed ambiguous associations, with BI partially correlating with dengue fever cases for two out of three years [42]. As in their earlier study at the same Cuban location [29], Sanchez et al [20] reported that while $BI_{max} \ge 4$ was a useful predictor for outbreaks at the block level, sensitivity during outbreaks ranged between 62% and 81.8% and specificity between 71.9% and 78.1%.

Use of Vector Indices as Transmission Thresholds

The Breteau Index (BI) was used as an outcome measure in seven studies [29–31,34,36–38] and BI_{max} threshold was considered in three (Table 2) [20,29,40]. Here, BI values ranged from 1 to 66 during periods when dengue transmission was recorded (Figure 2). In other studies, both recent [46] and historic [47], dengue transmission was recorded when BI values were lower than the widely accepted transmission threshold of 5. Notably, in a study in Trinidad, 'high' transmission (25–40 cases for 75% of sample 'cycles') took place in areas with relatively 'low' abundance (~BI<5) while, conversely, a consistently higher BI of 5.4 in neighbouring areas did not result in dengue cases [34]. In Rio de Janeiro, the BI did not correlate with dengue incidence and

transmission occurred in association with a wide range of BI levels (range 3.30–20.51) [38].

Discussion

With worldwide dengue transmission levels at an all time high, predicting dengue outbreaks in advance of their occurrence or identifying specific locations where outbreak risks are highest is of critical importance. This review considered the evidence that changes in vector populations can be correlated with dengue virus transmission and whether or not monitoring fluctuations in vector indices might be employed to provide reliable advance warning of impending dengue outbreaks.

Eighteen studies that had the potential to provide evidence of any association between vector indices and dengue incidence were identified and examined. Notably, only 4 studies utilized new data on both vector indices and dengue cases collected de novo as an integral part of the study. More common was a reliance on local government-level records for the dengue case data, a practice that potentially introduces error or bias for number of reasons. First, hospital reports are prone to selection bias, as asymptomatic/ inapparent infections may not be recorded and the actual number of cases may have been significantly underreported. Second, there can be a considerable delay between the times of onset of infection and reporting which, if the infection date is not calculated, would result in a temporal mismatch of vector and case data. Third, differences between the geographic location of the vector and dengue case data, or between the spatial units from which each was originally calculated, would result in a geographic mismatch or mask potential relationships, respectively.

The latter point is of particular significance not only from the point of view of these studies, but also when considering the design of future investigations. A growing body of evidence indicates that the distribution of dengue cases typically is highly clustered in both time and space. In various studies, post-dating those reviewed, the size of such clusters ranged from 800 m [48] to less than 100 m [49]. The effective area of such key 'pockets' or 'hotspots' is likely to be determined by dispersal of the vector [49,50] which itself can vary over time [51], and is influenced by house density [52] and by human movement within and beyond the infection cluster [53]. Consequently, in studies attempting to correlate vector indices with dengue transmission, and where the geographical unit is too large, high vector densities in key dengue hotspots might be diluted by inclusion of neighboring areas with low densities, thus masking any true relationships [see 38].

Indeed, human movement potentially confounds dengue vector data that derive from residential areas alone as increasingly, evidence indicates that only a proportion of dengue infections are transmitted in the individual's own home, with many infections (possibly the majority) resulting from bites by virus-infected mosquitoes at other houses, schools, workplaces or numerous locations remote from the home [53,54]. Clearly, this presents a serious challenge when considering the use of vector data for surveillance and highlights a need for inclusion of data from public locations [55] in addition to residential areas, in any surveillance program.

Returning to the studies examined in this review, the fact that there was no clear indication of any consistent association between vector indices and dengue cases is not unexpected, given the diverse and mostly weak study designs. One study found there was no apparent increase in vector indices coinciding with what was the largest increase in dengue fever cases of all areas studied [40], while in another, dengue transmission remained low despite exceptionally high vector indices [44]. In studies where correlations were calculated for HI, BI and dengue cases, regression coefficients ranged from weak/moderate non-significant (R = 0.43 and R = 0.35 respectively; p>0.05) [38], to moderate significant associations (R = 0.61 and R = 0.60 respectively, but only in the urban centre; p<0.05) [37].

Only two studies calculated pupal indices, even though fifteen of the eighteen studies reviewed were published more than three years after WHO acknowledged that the traditional *Stegomyia* indices were inadequate for the measurement of dengue vector abundance [56]. In the two studies included in this review that calculated pupal indices, only one reported increases in the pupal index, but its relationship with dengue cases was not statistically significant, possibly due to the low numbers of pupae recorded [30,35]. A major problem with pupal surveys is the difficulty in locating breeding sites and the potential existence of important or key but cryptic breeding sites (*e.g.* overhead tanks on houses or underground water reserves such as severs or wells) that may harbor significant proportions of the vector population [57,58].

Clearly, calculation of adult female *Aedes aegypti* indices is the most direct measure of exposure to dengue transmission [15]. Of the four studies reviewed that reported some correlation between vector indices and dengue cases, two [31,39] recorded adult vector data. The adult population of *Aedes aegypti* is rarely sampled, partly due to the erroneous but commonly held belief that carrying out such sampling is time-consuming, difficult or expensive [59].

Sampling adult female Aedes aegypti is a relatively simple task, though it can be limited by the fact that mosquito numbers often remain low during outbreaks [60]. Nonetheless, it is possible to aim to sample adult mosquitoes as a routine procedure with minimal additional training and resources. A number of novel sampling devices [61-63] offer the potential to monitor vectors during outbreaks [64] and at the spatial scale required to accurately sample populations of Ae. aegypti [65]. Simple affordable low-tech tools that enable localized sampling of adult Ae. aegypti and other mosquito vectors are available, with initial studies demonstrating their ease and effectiveness in comparison with older methods [66,67]. In Brazil, routine sampling of Ae. aegypti adults with gravid traps deployed at relatively low densities was used to identify high risk localities which were then targeted for vector control [68,69]. This 'Intelligent Dengue Monitoring' system was reported to have prevented over 27,000 dengue cases over two 'dengue seasons' between 2009 and 2011 with considerable reductions in cost burden to the communities where it was deployed [70].

None of the studies reported on viral infection rates in the vector. This perhaps is not surprising given that techniques suitable for application in routine surveillance, such as PCR or NS1, have not been available until recently, that vector infection rates with dengue virus are of the order of 1% even in areas where transmission is ongoing [64,70-72] and the cost of running the large numbers of tests to detect meaningful infection levels could be considered prohibitive for many authorities. Nonetheless, routine screening for dengue virus of trapped adult female Aedes *aegypti* is possible and has been incorporated into the routine surveillance program in Belo Horizonte, Brazil [73]. The relative low dispersal rates of Ae. aegypti as compared with the high mobility of humans as they commute daily from the home to the workplace, school, etc., means that virus-infection rates in the vector potentially could provide an accurate or epidemiologically valid indicator of dengue risk in any particular locality, thus informing vector control. Clearly, elucidating the relative value of such an index would require substantial research investment, while integrating it into routine surveillance programmes would demand significant sustained investment, but the importance of metrics like the sporozoite or entomological inoculation rates used in malaria epidemiology [59] already indicate the potential.

This review has also demonstrated the unreliability of accepted vector thresholds for dengue transmission. A number of studies reported dengue transmission at BI levels below the currently accepted threshold of 5 (Figure 2) [29,34,36-38] or when the HI was below 1% [74,75]. Elsewhere, Focks proposed a pupal productivity index of 0.25 as a threshold for dengue transmission in Honduras [76], yet in Brazil dengue transmission occurred at PPI levels of 0.15 [58]. While the desire for a single globally applicable transmission threshold is understandable, it seems unlikely that such a threshold exists, given the variety and complexity of other parameters that potentially influence the risk of outbreaks today [19,77,78]. Chadee concluded in 2009 that dengue transmission occurs, not at a fixed entomologic figure/ quantity but rather at a variable level based on numerous factors including seroprevalence, mosquito density and climate [30]. It is becoming increasingly apparent that thresholds differ at different locations and in different contexts, and while they must be calculated independently at each location [19,79]. Moreover, empirically defining thresholds, which must be expected to be dynamic, rising and falling as the susceptibility of the local population changes, will require comprehensive prospective, longitudinal vector studies [80], with simultaneous monitoring of the relationship between Ae. aegypti population densities and dengue virus transmission in a spatially relevant human cohort.

Study Limitations

In spite of reference searches and use of grey literature, publication bias will likely remain given the very nature of a systematic review. However, we also sought to further limit the effect of publication bias by placing no restriction on language, and those languages encountered were: English, French, Portuguese, Spanish and Chinese.

Additionally, one should be cautious when interpreting these data due to the study design of the 18 articles. As defined by QATAS assessment methods, study design was often weak (15 studies), meaning that studies were prone to bias and confounding factors, which may have skewed some of the reported associations. In addition, most (n = 13) studies relied on dengue case data from external sources, rather than obtaining study-generated data. With the exception of vector sampling and generation of vector index, there were few similarities in the approaches across the different studies.

Conclusions and Recommendations

Despite the widespread practice of collecting vector population data, the review has revealed that very few rigorous studies have been undertaken to determine the relationship between vector abundance and dengue transmission; of those that have been published, few provide tangible evidence of such a relationship, and therefore it is not possible to draw a firm conclusion. After decades of vector surveillance in many countries and considering the magnitude of the dengue threat today both in those and other countries that have recently experienced major dengue outbreaks, this is disappointing. Yet it is also indicative of the lack of basic knowledge of dengue epidemiology, in particular with regard to transmission. Clearly, this is a major knowledge gap that requires attention with a degree of urgency and the following research priorities are recommended:

• The relationship between vector population abundance and dengue transmission remains unknown and should be quantified. Studies should aim to collect new vector and

- The ideal and most powerful approach would be for a series of coordinated studies, to be carried out in multiple locations worldwide, as exemplified by recent examples [80]. To facilitate such studies, and ensure higher power in individual and combined datasets, the development of a standardized study design and protocols is a priority.
- Individual locations are also strongly encouraged to investigate the relationship independently. Many dengue-affected areas (cities, districts or similar spatial units) are likely to have substantial historic vector and dengue data that potentially may be suitable for appropriate analysis.
- Spatial heterogeneity and transmission at sites other than the home must be considered and carefully incorporated into any study design.
- The utilization of single global values of the Breteau (BI) or other vector indices as thresholds for dengue transmission is unreliable and is not recommended.
- While the need for a standardized reliable definition of a dengue outbreak has already been stated elsewhere [81], research into the relationship between vector abundance and dengue transmission should endeavor to develop a similar approach to defining reliable locality-specific vector population indices (*e.g.* thresholds, rates of increase, etc.) for use as early warning signals for impending increases in dengue transmission.
- Adoption of adult dengue vector sampling by all vector surveillance programs is urged. Various new trapping methods, as well as a simple resting catch approach, should be evaluated.
- Relationship between larval, pupal and adult stages of the vector population and the factors influencing adult emergence rates remain poorly understood. The paucity of fundamental knowledge of the ecology of mosquito vectors generally and the need for basic studies has been advocated elsewhere [82,83] and is true for *Ae. aegypti* and *Ae. albopictus*. A greater understanding of the ecology of dengue vectors is essential.

References

- Guzman MG, Halstead SB, Artsob H, Buchy P, Farrar J, et al. (2010) Dengue: a continuing global threat. Nature Publishing Group 8: S7–S16. doi:10.1038/ nrmicro2460.
- Simmons CP, Farrar JJ, Nguyen VVC, Wills B (2012) Dengue. N Engl J Med 366: 1423–1432. doi:10.1056/NEJMra1110265.
- Bhatt S, Gething PW, Brady OJ, Messina JP, Farlow AW, et al. (2013) The global distribution and burden of dengue. Nature 496: 504–507. doi:10.1038/ nature12060.
- World Health Organization (2012) Global strategy for dengue prevention and control 2012–2020. World Health Organization. 43 pp. Available: http://apps. who.int/iris/bitstream/10665/75303/1/9789241504034_eng.pdf
- Brady OJ, Gething PW, Bhatt S, Messina JP, Brownstein JS, et al. (2012) Refining the global spatial limits of dengue virus transmission by evidence-based consensus. PLoS Negl Trop Dis 6: e1760. doi:10.1371/journal.pntd.0001760.s009.
- Luz PM, Vanni T, Medlock J, Paltiel AD, Galvani AP (2011) Dengue vector control strategies in an urban setting: an economic modelling assessment. The Lancet 377: 1673–1680. Available: http://www.sciencedirect.com/science/ article/pii/S0140673611602468.
- Undurraga EA, Halasa YA, Shepard DS (2013) Use of expansion factors to estimate the burden of dengue in southeast Asia: A systematic analysis. PLoS Negl Trop Dis 7: e2056. doi:10.1371/journal.pntd.0002056.s003.
- Shepard DSD, Undurraga EAE, Halasa YAY (2013) Economic and disease burden of dengue in southeast Asia. PLoS Negl Trop Dis 7: e2055–e2055. doi:10.1371/journal.pntd.0002055.

In the absence of definitive evidence that dengue vector surveillance data can contribute to the prediction of dengue outbreaks, it might be tempting to consider abandoning the practice altogether. However, this would be a rash and premature judgment. At the very least, this systematic review has demonstrated that the potential of vector surveillance data has not yet been evaluated. Indeed, its full potential will not be apparent until its contribution to a complete predictive model incorporating all other covariates influencing dengue epidemiology have been considered. That will not be possible until multiple high quality studies investigating the relationship between vector populations and dengue transmission have been carried out.

Supporting Information

Checklist S1 PRISMA checklist. (DOC)

 Table S1
 Data extraction table summary for reviewed studies.

 (XLSX)

Table S2Assessment of the validity of reviewed studies: Table ofbias and QATQS (Quality Assessment Tool for QuantitativeStudies) rating for each study.(XLSX)

Acknowledgments

We thank David Valladares, Brian Suarez, Jacob Riveron, Carolina Rediguieri, Yang Wu and Walter Fabricio Silva Martins for assistance with English translations; Clare Strode, Paul Garner and Martin Chapman for advice on conducting systematic reviews; Brian Faragher for advice on statistical analyses. We also thank the reviewers of the submitted version for their constructive comments.

Author Contributions

Conceived and designed the experiments: LRB PJM. Analyzed the data: LRB. Wrote the paper: PJM LRB SRR. Contributed to the review protocol: LRB SRR PJM. Carried out literature searches: LRB. Reviewed the literature: LRB PJM. Wrote the conclusions and article drafts: PJM LRB. Drafted the manuscript: PJM LRB. Reviewed and approved the final manuscript: LRB SRR PJM.

- Halasa YA, Shepard DS, Zeng W (2012) Economic cost of dengue in Puerto Rico. Am J Trop Med Hyg 86: 745–752. doi:10.4269/ajmh.2012.11-0784.
- Martelli CMT, Nascimento NE, Suaya JA, Siqueira JB, Souza WV, et al. (2011) Quality of life among adults with confirmed dengue in Brazil. Am J Trop Med Hyg 85: 732–738. doi:10.4269/ajtmh.2011.11-0067.
- Lambrechts L, Scott TW, Gubler DJ (2010) Consequences of the expanding global distribution of *Aedes albopictus* for dengue virus transmission. PLoS Negl Trop Dis 4: e646. doi:10.1371/journal.pntd.0000646.t002.
- Barrera R, Amador M, MacKay AJ (2011) Population Dynamics of Aedes aegypti and Dengue as Influenced by Weather and Human Behavior in San Juan, Puerto Rico. PLoS Negl Trop Dis 5(12): e1378. doi:10.1371/ journal.pntd.0001378
- Campbell KM, Lin CD, Iamsirithaworn S, Scott TW (2013) The Complex Relationship between Weather and Dengue Virus Transmission in Thailand. Am J Trop Med Hyg 89: 1066–1080. doi:10.4269/ajtmh.13-0321.
- World Health Organization (2009) Dengue: Guidelines for diagnosis, treatment, prevention and control. WHO, Geneva, Switzerland: 1–160. Available: http:// www.ncbi.nlm.nih.gov/books/NBK143157
- Focks D (2004) A review of entomological sampling methods and indicators for dengue vectors. UNDP/World Bank/WHO Special Programme for Research and Training in Tropical Diseases: 1–40. Available: http://www.who.int/iris/ handle/10665/68575#sthash.OmTDGEBh.dpuf
- Connor ME, Monroe WM (1923) Stegomyia indices and their value in yellow fever control. Am J Trop Med Hyg 1: 9–19. Available: http://intl.ajtmh.org/ content/s1-3/1/9.full.pdf

- Focks D, Chadee DD (1997) Pupal survey: an epidemiologically significant surveillance method for *Aedes aegypti*: an example using data from Trinidad. Am J Trop Med Hyg 56: 159–167. Available: http://www.ncbi.nlm.nih.gov/ pubmed/9080874
- Tun-Lin W, Kay BH, Barnes A, Forsyth S (1996) Critical examination of Aedes aegypti indices: correlations with abundance. Am J Trop Med Hyg 54: 543–547. Available: http://www.ncbi.nlm.nih.gov/pubmed/8644913
- Kuno GG (1995) Review of the factors modulating dengue transmission. Epidemiol Rev 17: 321–335. Available: http://www.ncbi.nlm.nih.gov/ pubmed/8654514
- Sanchez L, Cortinas J, Peleaz O, Gutierrez H, Concepción D, et al. (2010) Breteau index threshold levels indicating risk for dengue transmission in areas with low *Aedes* infestation. Tropical Medicine & International Health 15: 173– 175. Available: http://www.ncbi.nlm.nih.gov/pubmed/20409286
- Scott TW, Morrison AC (2003) Aedes aegypti density and the risk of dengue virus transmission. In: Takken W, Scott TW (eds) Ecological aspects for application of genetically modified mosquitoes. FRONTIS, Dordrecht, The Netherlands, pp 187–206. Available: http://library.wur.nl/ojs/index.php/frontis/article/view/ 849
- 22. Focks D, Brenner RJ, Hayes J, Daniels E (2001) Transmission thresholds for dengue in terms of *Aedes aegypti* pupae per person with discussion of their utility in source reduction efforts. Am J Trop Med Hyg 62: 11–18. Available: http:// www.ajtmh.org/content/62/1/11.full.pdf#page = 1&view = FitH.
- Ellis AM, Garcia AJ, Focks D, Morrison AC, Scott TW (2011) Parameterization and sensitivity analysis of a complex simulation model for mosquito population dynamics, dengue transmission, and their control. Am J Trop Med Hyg 85: 257–264. doi:10.4269/ajtmh.2011.10-0516.
- Andraud M, Hens N, Marais C, Beutels P (2012) Dynamic epidemiological models for dengue transmission: A systematic review of structural approaches. PLoS ONE 7(11): e49085. doi:10.1371/journal.pone.0049085
- Badurdeen S, Valladares DB, Farrar J, Gozzer E, Kroeger A, et al. (2013) Sharing experiences: towards an evidence based model of dengue surveillance and outbreak response in Latin America and Asia. BMC Public Health 13: 1–1. doi:10.1186/1471-2458-13-607.
- Centre for Reviews and Dissemination (2009) Systematic Reviews: CRD's guidance for undertaking reviews in health care. York: University of York; 2009. Available: http://www.york.ac.uk/inst/crd/pdf/Systematic_Reviews.pdf
- Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA Group (2009) Preferred reporting items for systematic reviews and meta-analyses: The PRISMA Statement. Open Med 3: e123–e130. Available: http://www. plosmedicine.org/article/info%3Adoi%2F10.1371%2Fjournal.pmed.1000097
- National Collaborating Centre for Methods and Tools (2008) Quality assessment tool for quantitative studies. Hamilton, ON: McMaster University. (Updated 13 April, 2010). Retrieved from http://www.nccmt.ca/registry/view/eng/14.html.
- Sanchez L, Vanlerberghe V, Alfonso L, Marquetti MDC, Guzman MG, et al. (2006) Aedes aegypti larval indices and risk for dengue epidemics. Emerging Infect Dis 12: 800–806. Available: http://wwwnc.cdc.gov/eid/article/12/5/05-0866_ article.htm
- Chadee DD (2009) Dengue cases and Aedes aegypti indices in Trinidad, West Indies. Acta Trop 112: 174–180. doi:10.1016/j.actatropica.2009.07.017.
- 31. Pham HV, Doan HTM, Phan TTT, Minh NNT (2011) Ecological factors associated with dengue fever in a central highlands province, Vietnam. Stoch Environ Res Risk Assess 25: 485–494. Available: http://ovidsp.ovid.com/ ovidweb.cgi7T = JS&CSC = Y&NEWS = N&PAGE = fulltext&D = emed10&AN = 2011358286.
- 32. Gurtler RE, Garelli FM, Coto HD (2009) Effects of a five-year citywide intervention program to control Aedes aegypti and prevent dengue outbreaks in northern Argentina. PLoS Negl Trop Dis 3: e427. Available: http://ovidsp.ovid. com/ovidweb.cgi7T = JS&CSC = Y&NEWS = N&PAGE = fulltext&D = emed9&AN = 2009223964.
- Katyal R, Kumar K, Gil KS, Sharma RS (2003) Impact of intervention measures on DF/DHF cases and *Aedes aegypti* indices in Delhi, India: An update, 2001. Dengue Bulletin 27: 163–167. Available: http://repository.searo.who.int/ bitstream/123456789/15910/2/dbv27p163.pdf
- Chadee DD, Williams FL, Kitron UD (2005) Impact of vector control on a dengue fever outbreak in Trinidad, West Indies, in 1998. Tropical Medicine & International Health 10: 748–754. doi:10.1111/j.1365-3156.2005.01449.x.
- Romero-Vivas CME, Falconar AKI (2005) Investigation of relationships between *Aedes aegypti* egg, larvae, pupae, and adult density indices where their main breeding sites were located indoors. J Am Mosq Control Assoc 21: 15–21. Available: http://www.ncbi.nlm.nih.gov/pubmed/15825756
- Foo LC, Lim TW, Lee HL, Fang R (1985) Rainfall, abundance of *Aedes aegypti* and dengue infection in Selangor, Malaysia. The Southeast Asian Journal of Tropical Medicine and Public Health 16: 560–568. Available: http://www.ncbi. nlm.nih.gov/pubmed/3835698
- Sulaiman S, Pawanchee ZA, Arifin Z, Wahab A (1996) Relationship between Breteau and House indices and cases of dengue/dengue hemorrhagic fever in Kuala Lumpur, Malaysia. J Am Mosq Control Assoc 12: 494–496. Available: http://www.ncbi.nlm.nih.gov/pubmed/8887232
- Honorio NA, Nogueira RMR, Codeco CT, Carvalho MS, Cruz OG, et al. (2009) Spatial evaluation and modeling of dengue seroprevalence and vector density in Rio de Janeiro, Brazil. PLoS Negl Trop Dis 3(11): e545. doi: 10.1371/ journal.pntd.0000545

- 39. Rubio-Palis Y, Perez-Ybarra LM, Infante-Ruiz M, Comach G, Urdaneta-Marquez L (2011) Influence of climatic variables on dengue cases and abundance of Aedes aegypti (Diptera: Culicidae) in Maracay, Venezuela. Boletin De Malariologia Y Salud Ambiental 51: 145–157. Available: http://www.scielo. org.ve/scielo.ph?pid=S1690-46482011000200004&script=sci_abstract&thg=en
- 40. Lin CH, Wen TH (2011) Using geographically weighted regression (GWR) to explore spatial varying relationships of immature mosquitoes and human densities with the incidence of dengue. IJERPH 8: 2798–2815. Available: http://www.mdpi.com/1660-4601/8/7/2798
- 41. Chaikoolvatana A, Singhasivanon P, Haddawy P (2007) Utilization of a geographical information system for surveillance of *Aedes aegypti* and dengue haemorrhagic fever in north-eastern Thailand. Dengue Bulletin 31: 75–82. Available: http://repository.searo.who.int/handle/123456789/16080
- Chadee DD, Shivnauth B, Rawlins SC, Chen AA (2007) Climate, mosquito indices and the epidemiology of dengue fever in Trinidad (2002–2004). Ann Trop Med Parasitol 101: 69–77. doi:10.1179/136485907X157059.
- Correa PR, Franca E, Bogutchi TF (2005) Aedes aegypti infestation and occurrence of dengue in the city of Belo Horizonte, Brazil. Rev Saude Publica 39: 33–40. Available: http://www.ncbi.nlm.nih.gov/pubmed/15654458
- 44. Fernández R W, Lannacone O J, Rodríguez P E, Salazar C N, Valderrama R B, et al. (2005) Population behavior of *Aedes acypti* larvae for estimating dengue fever cases in Yurimaguas, Peru, 2000–2004. Rev Peru Med Exp Salud Publica 22: 175–182. Available: http://www.scielo.org.pe/scielo.php?pid=S1726-46342005000300004&script=sci_abstract&tlng = en
- Arboleda SS, Jaramillo-O NN, Peterson ATA (2012) Spatial and temporal dynamics of *Aedes aegypti* larval sites in Bello, Colombia. J Vector Ecol 37: 37–48. doi:10.1111/j.1948-7134.2012.00198.x.
- Hanna JN, Ritchie SA, Phillips DA, Serafin IL, Hills SL, et al. (2001) An epidemic of dengue 3 in far north Queensland, 1997–1999. Medical Journal of Australia 174: 178–182.
- Macdonald WW (1956) Aedes aegypti in Malaya. II. Larval and adult biology. Ann Trop Med Parasitol 50(4): 399–414.
- Vazquez-Prokopec GM, Kitron U, Montgomery B, Horne P, Ritchie SA (2010) Quantifying the spatial dimension of dengue virus epidemic spread within a tropical urban environment. PLoS Negl Trop Dis 4(12): e920. doi:10.1371/ journal.pntd.0000920
- Yoon I-K, Getis A, Aldstadt J, Rothman AL, Tannitisupawong D, et al. (2012) Fine scale spatiotemporal clustering of dengue virus transmission in children and *Aedes aegypti* in rural Thai villages. PLoS Negl Trop Dis 6(7): e1730. doi:10.1371/ journal.pntd.0001730
- Schafrick NH, Milbrath MO, Berrocal VJ, Wilson ML, Eisenberg JNS (2013) Spatial clustering of *Aedes aegypti* related to breeding container characteristics in coastal Ecuador: Implications for dengue control. Am J Trop Med Hyg 89: 758– 765
- 51. Duncombe J, Clements A, Davis J, Hu W, Weinstein P, et al. (2013) Spatiotemporal patterns of *Aedes aegypti* populations in Cairns, Australia: assessing drivers of dengue transmission. Tropical Medicine and International Health 18: 839–849. doi:10.1111/tmi.12115
- Kroeger A, Lenhart A, Ochoa M, Villegas E, Levy M, et al. (2006) Effective control of dengue vectors with curtains and water container covers treated with insecticide in Mexico and Venezuela: cluster randomised trials. BMJ 332: 1247– 1252
- Stoddard ST, Forshey BM, Morrison AC, Paz-Soldan VA, Vazquez-Prokopec GM, et al. (2013) House-to-house human movement drives dengue virus transmission. Proc Natl Acad Sci 110(3):994–9. doi: 10.1073/pnas.1213349110
- Stoddard ST, Morrison AC, Vazquez-Prokopec GM, Paz Soldan V, Kochel TJ, et al. (2009) The role of human movement in the transmission of vector-borne pathogens. PLoS Negl Trop Dis 3(7): e481. doi:10.1371/journal.pntd.0000481
- Morrison AC, Sihuincha M, Stancil JD, Zamora E, Astete H, et al. (2006) Aedes aegypti (Diptera: Culicidae) production from non-residential sites in the Amazonian city of Iquitos, Peru. Ann Trop Med Parasitol 100 Suppl 1:S73– S86. doi:10.1179/136485906X105534
- 56. World Health Organization (2000) Strengthening implementation of the global strategy for dengue fever/dengue haemorrhagic fever prevention and control. Report of the Informal Consultation 18–20 October 1999, Geneva WHO/ CDS/(DEN)/IC/2000.1 Available: http://www.who.int/csr/resources/ publications/dengue/whocdsdenic20001.pdf
- Barrera R, Amador M, Diaz A, Smith J, Munoz-Jordan JL, Rosario Y (2008) Unusual productivity of *Aedes aegypti* in septic tanks and its implications for dengue control. Med. Vet. Entomol 22: 62–69. doi: 10.1111/j.1365-2915.2008.00720.x.
- Pilger D, Lenhart A, Manrique-Saide P, Siqueira JB, Da Rocha WT, et al. (2011) Is routine dengue vector surveillance in central Brazil able to accurately monitor the *Aedes aegypti* population? Results from a pupal productivity survey. Tropical Medicine & International Health 16: 1143–1150. doi:10.1111/j.1365-3156.2011.02818.x.
- Anders KL, Hay SI (2012) Lessons from malaria control to help meet the rising challenge of dengue. Lancet Infec Dis 12(12) 977–984. doi: 10.1016/S1473-3099(12)70246-3.
- Goh KT (1997) Dengue–a re-emerging infectious disease in Singapore. Ann Acad Med Singap 26: 664–670.
- 61. Maciel-de-Freitas R, Peres RC, Alves F, Brandolini MB (2008) Mosquito traps designed to capture *Aedes aegypti* (Diptera: Culicidae) females: preliminary comparison of Adultrap, MosquiTRAP and backpack aspirator efficiency in a dengue-endemic area of Brazil. Mem Inst Oswaldo Cruz 103: 602–5.

- Barrera R, Mackay AJ, Amador M (2013) An improved autocidal gravid ovitrap for the control and surveillance of *Aedes aegypti*. Parasit Vectors 6: 225. doi: 10.1186/1756-3305-6-225.
- Ritchie SA, Buhagiar TS, Townsend M, Hoffmann A, Van Den Hurk AF, McMahon JL, Eiras AE. (2014) Field validation of the gravid *Aedes* trap (GAT) for collection of *Aedes aegypti* (Diptera: Culicidae). J Med Entomol 51: 210–9.
- Ritchie SA, Pyke AT, Hall-Mendelin S, Day A, Mores CN, et al. (2013) An explosive epidemic of DENV-3 in Cairns, Australia. PLoS ONE 8(7): e68137. doi:10.1371/journal.pone.0068137.
- Barrera R (2011) Spatial stability of adult Aedes aegypti populations. Am J Trop Med Hyg, 85:1087–1092 doi:10.4269/ajtmh.2011.11-0381.
- 66. Maia MF, Robinson A, John A, Mgando J, Simfukwe E, Moore SJ (2011) Comparison of the CDC Backpack aspirator and the Prokopack aspirator for sampling indoor- and outdoor-resting mosquitoes in southern Tanzania. Parasites & Vectors 4:124. Available: http://www.parasitesandvectors.com/ content/4/1/124
- Vazquez-Prokopec GM, William A. Galvin WA, Kelly R, Kitron U (2009) A new, cost-effective, battery-powered aspirator for adult mosquito collections. J Med Entomol 46: 1256–1259. Available: http://www.ncbi.nlm.nih.gov/ pubmed/19960668
- Eiras AE, Resende MC (2009) Preliminary evaluation of the 'Dengue-MI' technology for Aedes aegypti monitoring and control. Cad Saude Publica 25 Suppl 1:S45–58. doi: 10.1590/S0102-311X2009001300005.
- Pepin KM, Marques-Toledo C, Scherer L, Morais MM, Ellis B, et al. (2013) Cost-effectiveness of novel System of mosquito surveillance and control, Brazil. Emerging Infect Dis 19(4) 542. doi 10.3201/eid1904.120117.
- Mammen MP Jr, Pimgate C, Koenraadt CJM, Rothman AL, Aldstadt J, et al. (2008) Spatial and temporal clustering of dengue virus transmission in Thai villages. PLoS Med 5(11): e205. doi:10.1371/journal. pmed.0050205.
- Garcia-Rejon J, Lorono-Pino MA, Farfan-Ale JA, Flores-Flores L, Rosado-Paredes E, et al. (2008) Dengue virus-infected Aedes aegypti in the home environment. Am J Trop Med Hyg 79: 940–950. Available: http://www.ajtmh. org/content/79/6/940.long.
- Yoon I-K, Getis A, Aldstadt J, Rothman AL, Tannitisupawong D, et al. (2012) Fine scale spatiotemporal clustering of dengue virus transmission in children and Aedes aegypti in rural Thai villages. PLoS Negl Trop Dis 6(7): e1730. doi:10.1371/journal.pntd.0001730.
- 73. de Figueiredo RMP, Gomes Mourão MP, Conte Abi-Abib YE, de Oliveira CM, Roque R, de Azara T, Ohly J, Degener C, Geier M, Eiras AE (2013) Identification of dengue viruses in naturally infected *Aedes aegypti* females captured with BioGents (BG)- Sentinel traps in Manaus, Amazonas, Brazil.

Revista da Sociedade Brasileira de Medicina Tropical 46:221–222. http://dx. doi.org/10.1590/0037-8682-1096-2013

- 74. Goh KT, Ng SK, Chan YC, Lim SJ, Chua EC (1987) Epidemiological aspects of an outbreak of dengue fever/dengue haemorrhagic fever in Singapore. The Southeast Asian Journal of Tropical Medicine and Public Health 18: 295–302. Available: http://www.ncbi.nlm.nih.gov/pubmed/3433161
- 75. Koh BKW, Ng LC, Kita Y, Tang CS, Ang LW, et al. (2008) The 2005 dengue epidemic in Singapore: epidemiology, prevention and control. Ann Acad Med Singap 37: 538–545. Available: http://www.google.co.uk/scarch?client = safari&rks = enus&q = The+2005+dengue+epidemic+in+Singapore+epidemiology+prevention+ and+control&ie = UTF-8&oe = UTF-8&redir_esc = &ei = Uj2IT8_UG8G_0Wc1um7CQ.
- Focks DA, Daniels E, Haile DG & Keesling JE (1995) A simulation model of the epidemiology of urban dengue fever: literature analysis, model development, preliminary validation, and samples of simulation results. American Journal of Tropical Medicine and Hygiene 53: 489–506.
- Reich NG, Shrestha S, King AA, Rohani P, Lessler J, et al. (2013) Interactions between serotypes of dengue highlight epidemiological impact of crossimmunity. Journal of The Royal Society Interface 10: 86. doi:10.3201/ eid1302.060539.
- Rabaa MA, Simmons CP, Fox A, Le MQ, Nguyen TTT, et al. (2013) Dengue virus in sub-tropical northern and central Viet Nam: Population immunity and climate shape patterns of viral invasion and maintenance. PLoS Negl Trop Dis 7(12): e2581. doi:10.1371/journal.pntd.0002581
- Sommerfeld J, Kroeger A (2012) Eco-bio-social research on dengue in Asia: a multicountry study on ecosystem and community-based approaches for the control of dengue vectors in urban and peri-urban Asia. Pathog Glob Health 106: 428–35. doi: 10.1179/2047773212Y.0000000055.
- Morrison AC1, Gray K, Getis A, Astete H, Sihuincha M, Focks D, Watts D, Stancil JD, Olson JG, Blair P, Scott TW. (2004) Temporal and geographic patterns of *Aedes aegypti* (Diptera: Culicidae) production in Iquitos, Peru. J Med Entomol 41:1123–42.
- Runge-Ranzinger S, Horstick O, Marx M, Kroeger A (2008) What does dengue disease surveillance contribute to predicting and detecting outbreaks and describing trends? Tropical Medicine & International Health 13: 1022–1041. doi:10.1111/j.1365-3156.2008.02112.x.
- Ferguson HM, Dornhaus A, Beeche A, Borgemeister C, Gottlieb M, et al. (2010) Ecology: a prerequisite for malaria elimination and eradication. PLoS Med 7:e1000303. doi: 10.1371/journal.pmed.1000303.
- Godfray HC (2013) Mosquito ecology and control of malaria. J Anim Ecol 82(1): 15–25. doi: 10.1111/1365-2656.12003.